Are Busy Boards Effective Monitors?

ELIEZER M. FICH and ANIL SHIVDASANI*

ABSTRACT

Firms with busy boards, those in which a majority of outside directors hold three or more directorships, are associated with weak corporate governance. These firms exhibit lower market-to-book ratios, weaker profitability, and lower sensitivity of CEO turnover to firm performance. Independent but busy boards display CEO turnover-performance sensitivities indistinguishable from those of inside-dominated boards. Departures of busy outside directors generate positive abnormal returns (ARs). When directors become busy as a result of acquiring an additional directorship, other companies in which they hold board seats experience negative ARs. Busy outside directors are more likely to depart boards following poor performance.

On December 28, 2000, the Wall Street Journal reported that Elaine L. Chao would be a nominee for President-elect George W. Bush's cabinet. Only a few days prior to Ms. Chao's confirmation as labor secretary, another Journal article described a growing trend among firms to limit the number of board seats their directors sit on because serving on too many boards may be detrimental to the quality of corporate governance. Coincidentally, this article also featured Ms. Chao as one of the 10 busiest directors among large U.S. corporations. As expected, upon her cabinet confirmation, Ms. Chao resigned her directorships at C.R. Bard, Clorox, Columbia/HCA Healthcare, Dole Foods, Northwest Airlines, and Protective Life.

Ms. Chao's cabinet appointment permits a case study analysis of the increasingly popular notion among shareholder activists, institutional investors, regulators, and many corporations that serving on several boards causes directors to be busy, rendering them ineffective monitors of corporate management. Using standard event study methodology, we find that Ms. Chao's impending departure from the six boards in which she served as an outside director was viewed enthusiastically by investors. Table I shows that the mean 2-day cumulative

*Fich is with Drexel University and Shivdasani is with the University of North Carolina at Chapel Hill. The paper benefited from comments by participants at the 2005 American Finance Association meetings, the 2004 Financial Research Association conference, and by seminar participants at Drexel, INSEAD, Seton Hall, North Carolina State, University of North Carolina, and Universidade Catolica de Portugal. The authors thank Anup Agrawal, Stuart Gillan, Bill Greene, Naveen Khanna, Robert Stambaugh, David Yermack, and an anonymous referee for helpful suggestions. The authors acknowledge financial support from the Wachovia Center for Corporate Finance.

¹ Cummings, Jeanne, and Greg Jaffe (2000), A floated name for cabinet lands with a thud, *Wall Street Journal*, Eastern Edition, December 28, A12.

² Lublin, Joann S. (2001), Multiple seats of power—Companies are cracking down on number of directorships board members can hold, *Wall Street Journal*, January 23, B1.

Table I Investor Reaction to Elaine Chao's Cabinet Nomination

Two-day cumulative abnormal returns (CARs) for the firms in which Elaine L. Chao served as an outside director. CARs are computed for all firms around December 28, 2000 (day "0"), the day when the Wall Street Journal first announced that Elaine L. Chao would join President-elect George W. Bush's cabinet. The sample includes the following firms: C.R. Bard, Clorox, Columbia/HCA Healthcare, Dole Food, Northwest Airlines, and Protective Life. Following her confirmation as secretary of labor, Ms. Chao resigned her directorships in these firms. We report t-statistics and Wilcoxon rank Z-statistics using a two-tailed test for significance.

Returns	Days	N	Mean Return	t-Statistic	Positive: Negative	Median Return	Wilcoxon Z
Raw return Cumulative abnormal return (CAR)	(- 1 - /	6 6	5.24% $3.80%$	1.99 2.22	6:0 6:0	4.49% $3.05%$	2.22 1.81

abnormal return (CAR) is 3.8% (*t*-statistic = 2.2) and the median CAR is 3.05% (Wilcoxon Z=1.8). All six firms in the study elicit positive investor reactions at announcement.³

While illustrative, this case study evidence is subject to a number of caveats. Investors might expect the six firms, whose boards Ms. Chao vacated, to benefit from her new political influence. Investors may also reassess the quality of the remaining board members due to a "halo effect" surrounding her nomination. Even if the stock price effect reflects the departure of a busy director, it is likely that the magnitude of this effect is exaggerated due to her status as one of America's busiest directors. Nonetheless, the evidence is suggestive of a negative impact of busy directors on firm value. Whether this effect holds in a systematic fashion across a broad sample of firms is the focus of this paper.

There is a growing literature that shows that serving on multiple boards can be a source of both valuable experience and reputational benefits for outside directors. Fama and Jensen (1983) note that reputational effects can be important incentives for outside directors. However, there is comparatively little evidence on the costs associated with serving on multiple boards, and the prior research on this topic is inconclusive. Beasley (1996) reports that the probability of committing accounting fraud is positively related to the average number of directorships held by outside directors. Core, Holthausen, and Larcker (1999) report that busy directors set excessively high levels of CEO compensation, which in turn leads to poor firm performance. In contrast, Ferris, Jagannathan, and Pritchard (2003) find no relation between the average number of directorships held by outside directors and the firm's market-to-book ratio.

We extend this literature along several dimensions. We show that inferences on whether multiple board seats held by directors affect firm performance are

³ We check for whether other news events might explain the observed abnormal returns. However, a Lexis-Nexis search around the announcement date fails to uncover release of other significant corporate news.

sensitive to how one identifies busy directors. Using measures of the fraction of outside directors that are busy, we find that busy boards display patterns associated with weaker corporate governance. Our findings differ from those reported by Ferris et al. (2003), who claim that busy boards are as effective as nonbusy boards at monitoring. We argue that their methodological choices and econometric specification lead to low statistical power for detecting the relation that we document between performance and busy outside directors.

Our base case results use the market-to-book ratio as a measure of firm performance. Given the long tradition of using the market-to-book ratio in this context, this specification enables us to directly compare our findings with prior studies. However, we are sensitive to the concern that market-to-book ratio models may be misspecified, since this measure has a number of alternative interpretations. As a proxy for a firm's marginal Q ratio, market-to-book ratio also measures a firm's incentive to invest. In addition, this ratio is also used as a systematic risk factor (Fama and French (1992)). To alleviate these concerns, we supplement our analysis with a number of additional tests that are relatively immune to the specification issues that arise in regressions using the market-to-book ratio.

Our results show that firms in which a majority of outside directors hold three or more board seats have significantly lower market-to-book ratios than firms in which a majority of outside directors hold fewer than three board seats; the magnitude of this effect is economically meaningful. The negative relation between market-to-book ratios and busy outside directors is robust to a wide range of sensitivity tests. We conduct tests to examine the potential endogeneity of busy outside directors with respect to firm performance. Using data on director appointments and departures, we are unable to detect any pattern indicating that poor firm performance influences board composition in a manner that causes a board's outside directors to become more busy.

As an alternative to using market-to-book ratios, we examine the effect of busy boards on measures of accounting performance. Using panel data regressions, we also find that an inverse relation holds between several accounting-based measures of operating performance and a majority of busy outside directors on the board.

Additional evidence that boards dominated by busy outside directors contribute to weaker corporate governance comes from an analysis of forced CEO turnover in our sample. We show that boards in which the majority of outside directors hold three or more directorships are less likely to remove a CEO for poor performance. Consistent with prior research, we find that outside-dominated boards are more likely to remove CEOs for poor performance than inside-dominated boards. However, our results suggest that a significant relation between turnover and performance holds only when a majority of outside directors on the board are not regarded as busy. Our tests reveal that forced CEO turnover is insensitive to firm performance when the majority of outside directors are busy, even if the board is dominated by outside directors.

Therefore, the extent to which outside directors are busy appears to be an important determinant of the effectiveness of outside-dominated boards in corporate governance.

Another piece of evidence comes from analysis of announcements of outside director departures in our broader sample. Similar to the case study evidence for Ms. Chao, abnormal returns (ARs) related to departure announcements of busy outside directors are significantly positive. Indeed, these returns are significantly higher than the ARs for departures of nonbusy outside directors. In addition, the results also indicate that departures of busy outside directors are viewed particularly favorably when a majority of the remaining outside directors on the board is not busy. Finally, we examine how stock prices respond when an incumbent director acquires an additional board seat. We find that when directors become busy as a result of obtaining a new board seat, stock prices tend to drop for the firms in which they are incumbent directors. Moreover, we also find that the decrease in stock price tends to be greater for firms in which the appointment causes the majority of the board's outside directors to be reclassified as busy.

Collectively, our results indicate that when a majority of outside directors are busy, firm performance suffers. At the same time, substantial evidence from prior studies suggests that the number of board seats held by directors is related to their performance as monitors and is correlated to their reputational capital. Ferris et al. (2003) find that the first appointment of a busy director to a board is good news for shareholders, implying that the enhanced experience or reputation of such directors is beneficial. However, our results suggest that there is also a cost to holding numerous board seats. As the number of outside directors sitting on multiple boards increases, boards are inclined to become distracted and monitoring intensity is likely to suffer. Therefore, our results imply that it may not be optimal for firms to select directors primarily based on the number of other boards they sit on, since this may lead to an overcommitted board.

Our results should not be interpreted as endorsing the recent efforts of institutional investors and corporate governance policy advocates in curbing the directorships held by outside directors for at least two reasons. First, there is substantial evidence that outside directorships tend to be correlated with a director's reputational capital and that the market for outside directorships provides an important source of incentives for outside directors to serve as monitors. Therefore, attempts to limit the number of outside directorships may reduce the strength of the incentives for some outside directors to engage in effective corporate governance. Second, our results relate primarily to the costs faced by firms that appoint busy outside directors—we are silent on the benefits that appointing companies might obtain when their executives join other boards as outside directors. Recent work by Perry and Peyer (2005) shows that sending firms benefit when their executives receive additional directorships, if measures of agency costs in these firms are relatively low. While our paper points to the potential benefits of limiting the number of board seats held by outside directors, policy recommendations on this issue should also incorporate the expected costs of curtailed director incentives and those borne by sending firms.

Our paper proceeds as follows. Section I reviews the relevant literature and formulates our research questions. Section II describes our sample. Section III studies whether busy boards affect firm performance. Section IV details our empirical tests on appointments and departures of outside directors. Section V investigates whether busy boards play a role during events of CEO turnover. Section VI analyzes investor reactions related to the departure of busy outside directors and also provides evidence on the impact of additional board seats on firms in which the director serves as an incumbent outside director. Section VII concludes.

I. Prior Literature on Directorships

Fama (1980) and Fama and Jensen (1983) argue that the market for outside directorships serves as an important source of incentives for outside directors to develop reputations as monitoring specialists. Mace (1986) suggests that outside directorships are perceived to be valuable because they provide executives with prestige, visibility, and commercial contacts.

Support for the reputational capital view of directorships comes from several studies which show that the number of boards that outside directors sit on is tied to the performance of the firms in which these directors are incumbents, either as CEOs or as outside directors. This pattern is documented for financially distressed companies (Gilson (1990)), for firms that cut dividends (Kaplan and Reishus (1990)) and opt out of stringent state antitakeover provisions (Coles and Hoi (2003)), for companies that fire their CEOs (Farrell and Whidbee (2000)), for firms that are sold (Harford (2003)), for CEOs following retirement (Brickley, Linck, and Coles (1999)), as well as for broad samples of firms (Yermack (2004)). Accordingly, several studies use the number of board seats held by an outside director as a proxy for the director's reputation in the external labor market (Shivdasani (1993), Vafeas (1999), Brown and Maloney (1999)).

While the number of directorships appears to be closely linked to directors' reputational capital, other studies suggest that too many directorships may lower the effectiveness of outside directors as corporate monitors (see, e.g., Core et al. (1999), Shivdasani and Yermack (1999)). Core et al. (1999) find that busy outside directors provide CEOs with excessive compensation packages, which in turn leads to weaker firm performance. Consistent with such a view, the National Association of Corporate Directors and the Council for Institutional Investors have adopted resolutions calling for limits on the number of directorships held by directors of publicly traded companies.⁴

⁴ See the Report of the National Association of Corporate Directors Blue Ribbon Commission on Director Professionalism (1996), and the Core Policies, Positions and Explanatory Notes from the Council of Institutional Investors (1998).

Ferris et al. (2003) test whether multiple board appointments by directors harm firm performance. They fail to detect any evidence of a systematic relation between the market-to-book ratio and the average number of board seats held by directors; they conclude that proposals calling for limits on multiple board appointments are misguided. However, several aspects of their research design prevent them from detecting the relation that we document between multiple directorships and firm performance.

First, market-to-book ratio can measure both the value added by management as well as the value of intangible assets such as future investment opportunities. Ferris et al. (2003) estimate cross-sectional regressions of the market-to-book ratio on director attributes but their regressions do not control for growth opportunities, which confounds the interpretation of their results.

Second, unlike Ferris et al. (2003) who estimate a cross-sectional model using 1995 data, we analyze panel data using fixed effects regressions. The fixed effects approach is robust to the presence of omitted firm-specific variables that would lead to biased estimates in an ordinary least squares (OLS) framework. Given the high correlation between the market-to-book ratio and corporate governance variables with numerous other company attributes, we view the fixed effects framework as offering significantly more reliable estimates than OLS regressions.⁵

A third distinction between our paper and Ferris et al.'s (2003) is in the identification of busy boards. They employ four measures to capture busy boards—three of these focus on directorships held by both inside and outside directors, while only one relates specifically to outside directors. To measure busy outside directors, they calculate the average number of board seats held by outside directors. Our variables, however, focus exclusively on whether outside directors are busy under the premise that inside and gray directors sit on the board for reasons other than the monitoring of management. Further, as we describe below, there is wide dispersion in the number of board seats held by outside directors, thereby making the average number of directorships a noisy measure of whether outside directors as a group are busy. We therefore employ an alternative metric that treats boards as busy if a majority of the outside directors sit on three or more boards.

Our paper is complementary to recent work by Perry and Peyer (2005) who examine announcement effects of outside director appointments for sending firms. They find that when executives join other boards as outside directors, the announcement return for the sending firm is positive when the executive has high stock ownership or the firm has an independent board. They argue that when executives have strong incentives to enhance shareholder value, accumulation of board seats by these executives has a positive impact on firm value.

 $^{^5}$ Ferris et al. (2003) also use the average return on assets (ROA) over 1993 to 1995 as a measure of performance. As with their market-to-book ratio regressions, the ROA specifications do not control for firm-specific effects.

In sum, there is substantial evidence supporting the view that outside directorships serve as a measure of a director's reputational capital. However, there is disagreement on whether sitting on numerous boards detracts from the ability of outside directors to perform as effective monitors. Our tests are designed to address the question of whether directors that serve on numerous boards tend to contribute to weaker corporate governance at these firms.

II. Sample and Data

A. Sample Selection

Our sample consists of firms that appear in the 1992 Forbes 500 lists of largest corporations based on assets, sales, market capitalization, or net income during the 7-year period from 1989 to 1995. We impose three screening criteria. First, we require that each company in the sample has at least 2 consecutive years of financial data available from the Center for Research in Security Prices and from Compustat. Second, relevant Securities and Exchange Commission filings have to be available on the Edgar data retrieval system. Third, utility and financial companies are excluded from the sample since regulatory effects may lead to a more limited role for their boards of directors. These criteria yield a final sample of 3,366 observations for 508 industrial companies across the 7 years.

For each firm, we collect data on corporate governance variables from proxy statements filed for each company during the sample period. Each director is classified according to his/her principal occupation. Full-time employees of the firm are designated as insiders. Directors associated with the company, former employees, those with existing family or commercial ties with the firm other than their directorship, or those with interlocking directorships with the CEO are designated as "gray." Directors that do not fit the description for inside or gray directors are classified as outside directors. We categorize boards as being interlocked if the CEO sits on the board of an outside director.

Descriptive statistics for key variables for the 508 companies are presented in Panel A of Table II. On average, outside directors hold 3.11 directorships (the median is 2.89). We count directorships held in all publicly traded firms but do not consider directorships held in nonpublic firms, not-for-profit and charitable organizations, trusts, and associations.

We consider outside directors busy if they serve on three or more boards. Although the three-directorship criterion is admittedly somewhat arbitrary, we choose this cutoff for several reasons. First, the mean and median number of directorships in the sample is close to three, resulting in a roughly even split between busy and nonbusy outside directors. Second, it reflects the recommendation by the Council for Institutional Investors that directors should sit on no more than two boards. Finally, our definition is consistent with prior work by Core et al. (1999) and Ferris et al. (2003) who also use the three-directorship benchmark for classifying executives as busy.

Using this definition, 52% of the outside directors in the sample are classified as busy. Perry and Peyer (2005) report a comparable frequency of busy outside directors in their sample. To measure the prevalence of busy outside directors on the board, we construct a (0, 1) indicator that takes the value of one if 50% or

Table II Data Description

Panel A provides descriptive statistics for characteristics of our sample firms. The sample consists of 3,366 annual observations for 508 companies between 1989 and 1995. Companies are included in the sample if they are listed by *Forbes* magazine as one of the largest U.S. public corporations in its 1992 survey of the 500 largest U.S. public companies in any of the categories of market capitalization, sales, net income, or assets. The sample excludes private, utility, and financial companies. The table presents the mean, median, and *SD* for each variable, as well as the Spearman sample correlation coefficient between all variables and a (0, 1) indicator that equals one if is the board is defined as busy, which occurs when 50% or more of the board's outside directors hold three or more directorships. *, ***, and *** denote statistical significance at the 1%, 5%, and 10% levels, respectively. Panel B shows characteristics of 2,314 outside directors appointed to the boards of our sample firms from 1989 to 1995. Outside directors are those that are not current or former employees of the firm, are not relatives of the CEO, have no business deals with the firm other than their directorship, and do not have interlocking directorships with the CEO. We classify boards as being interlocked if the CEO sits on the board of an outside director. Data on director characteristics are obtained from annual proxy statements.

	Panel A			
Variable	Mean	Median	SD	Correlation with "Busy Board"
Board Characteristics				_
Directorships per outside director	3.11	2.89	2.23	0.22^{*}
Percentage of inside directors	29.67	26.05	15.03	-0.07^{***}
Percentage of gray directors	15.02	9.21	13.32	-0.12***
Percentage of outside directors	55.33	56.23	17.12	0.68^{*}
Percentage of directors who are other firms' CEOs	14.96	13.20	11.70	0.56^{*}
Percentage of busy directors	52.26	_	_	_
Percentage of busy boards	21.42	_	-	-
Board size	11.88	12	2.95	0.15^{*}
Presence of interlocked board	0.36	0	0.72	0.48**
Directors' fees (1995 dollars)	35,904	27,601	13,562	0.29^{*}
Number of board meetings/year	7.56	7	2.56	0.31^{*}
Governance Structure				
CEO from founding family (0, 1)	0.26	0	0.39	-0.28***
Non-CEO chairman of board (0, 1)	0.15	0	0.33	-0.08***
CEO's tenure as CEO	8.68	7.5	7.68	0.12^{**}
CEO's age	58.06	56	7.04	0.00
Insider ownership (% common)	6.97	2.22	13.67	-0.21**
Institutional ownership (% common)	49.13	33.33	13.92	-0.06*
Firm Characteristics				
Total sales (1995 \$MM)	9,016.01	3,444.72	21,100.23	0.31^{*}
EBIT/Total assets	0.191	0.150	0.128	0.10^{*}
Firm age (years since incorporation)	23.6	12	9.33	0.45^{*}

(continued)

Table II—Continued

Panel B			
	Mean	Median	SD
Directorships per director	3.04	2.00	1.99
Percentage of appointees with three or more directorships	17.11	_	_
Age of the appointee	57	55	3.82
Equity ownership appointee (% of common)	0.03	0.07	0.15
Percentage of appointees that represent a board expansion	33.03	_	_
Percentage of appointees that replace an independent director	52.23	_	_
Percentage of appointees that replace an inside director	9.81	_	_
Percentage of appointees that replace a gray director	4.92	_	_
Percentage of appointees without prior board experience	13.56	_	_
Percentage of appointees who are commercial or investment bankers	7.02	_	_
Percentage of appointees who are current Forbes 500 executives	20.04	-	_
Percentage of appointees who are current CEOs of other firms	42.05	-	_
Percentage of appointees who are retired CEOs of other firms	18.12	_	-

more of the board's outside directors are busy. Throughout the paper, we refer to this variable as the "busy board" indicator. Panel A shows that 21% of the firms in the sample have busy boards.

A typical board has approximately 12 directors, 55.33% of whom are outsiders. The average board meets just under eight times a year. In Table II, we present the correlation of certain firm characteristics with the busy board indicator. This variable exhibits a positive correlation with the average directorships held by outside directors, the presence of an interlocking board, director fees, the frequency of board meetings, firm age, operating profit margin, and total sales. We observe a negative correlation between busy board and the percentage of inside and gray directors, ownership by insiders, and CEOs from founding families.

We track annual appointments of outside directors to the boards of the 508 firms during the 7-year period. Panel B of Table II presents key characteristics for the 2,314 individuals who are appointed as independent directors to the boards of these companies. A typical outsider is in her mid-50s and owns very little equity in the other boards on which she serves. Most of the appointees (52%) replace another independent director. These characteristics are comparable to those reported by Shivdasani and Yermack (1999) who study director appointments between 1994 and 1996. About 20% of all outside directors are current *Forbes* 500 executives, and almost 14% have no prior board experience. This last statistic is comparable to that reported by Ferris et al. (2003) who study director data for firms during the 1995 proxy season.

B. Average Directorships versus Busy Boards

Understanding what constitutes a busy board is a central issue underlying our tests. We consider boards busy if 50% or more of the outside directors hold three or more board seats instead of using the average number of directorships by outside directors to identify busy boards. At issue is the extreme skewness in the distribution of board seats held by outside directors. An example is helpful in illustrating this measurement issue.

Panels A through D of Table III report board appointments held by outside directors at Host Marriott, Gannett Newspapers, The Clorox Company, and MGM Grand, Inc., as disclosed in their 1993 proxy statements. While the ratio of total directorships to outside directors for Host Marriott and Gannett Newspapers is similar, 3.5 and 3.4, respectively, we determine that Host Marriott has a busy board, but not Gannett Newspapers. Conversely, a comparison of MGM Grand and Clorox demonstrates that a high average number of directorships does not necessarily indicate that a majority of outside directors are busy. The average ratio of directorships by outside directors is 3.66 for MGM Grand and only 2.66 for Clorox. However, 50% of the outside directors at Clorox are busy as compared to only 33% at MGM.

Panel E of Table III shows that a one-to-one correspondence between the average number of directorships and busy boards also fails to hold in the full sample. We divide the sample into four groups based on the percentage of outside directors that are classified as busy. When more than 75% of the outside directors are busy, the average number of directorships per outside director is 3.35. However, when only 25–50% of the outside directors are busy, the average number of directorships held by outside directors is 3.41. Our measurement treats boards in the first group as busy, while Ferris et al. (2003) would treat firms in the second group as having busier boards. As we illustrate later, our measurement appears to highlight a stronger link between busy boards and firm performance than using the average number of board seats variable.

III. Busy Boards and Firm Performance

Our first set of tests involves panel data estimates relating the market-to-book ratio to busy boards and other corporate governance and financial attributes. These models assume that a high market-to-book ratio is indicative of good management and governance. However, alternative interpretations of a high market-to-book ratio are equally plausible. In particular, if financial or liquidity constraints cause some firms to underinvest, the potential value of unexploited investments may lead to a high marginal Tobin's Q. If underinvestment is pervasive, our formulation would erroneously treat a high market-to-book ratio as indicative of good governance. We address this issue by using a number of controls for investment opportunities. However, we recognize that all measures are subject to measurement error. Therefore, we also supplement the market-to-book ratio tests with similar models estimating operating performance. Since historical operating performance does not employ market prices,

Table III Directorships by Outside Directors

Panels A through D report the total number of directorships held by outside directors, the mean directorships per outside director, and the percentage of outside directors holding three or more directorships for four companies in our data set during the 1993 proxy season. The total number of directorships simply counts the number of total boards of publicly traded firms on which the outside director serves. We do not count board service in private firms, charitable institutions, or not-for-profit organizations. The last row in each panel provides a (0,1) variable for whether boards are busy. We code boards as busy, with a one, if 50% or more outside directors hold three or more total directorships. Panel E reports mean directorships per outside director and per board for our sample firms according to the percentage of outside directors holding three or more directorships. Directorships per outside director are estimated as the total directorships held by outside directors divided by the number of outside directors. Similarly, directorships per board are all directorships held by every director, regardless of his/her classification, divided by board size.

Director	Main Occupation	Total Directorships
	Panel A: Host Marriott—Outside Directors 1993	
R. T. Ammon	Former Partner, Kohlberg Kravis Roberts & Co.	4
A. D. McLaughlin	President, Federal City Council (former U.S. secretary of labor)	8
H. L. Vincent, Jr.	Retired Vice-Chairman, Booz-Allen & Hamilton	1
A. J. Young	Vice Chairman, Law Companies Group, Inc.	1
	Total directorships	14
	Total directorships/outside directors	14/4 = 3.5
	Percentage with three or more directorships	50%
	Is the board busy? $(0 = No, 1 = Yes)$	Yes
	Panel B: Gannett Newspapers—Outside Directors 1993	1
Rosalyn Carter	Former First Lady of the United States of America	1
C. T. Rowan	President, CTR Productions	2
D. D. Wharton	CEO, Fund for Corporate Initiatives	3
A. F. Brimmer	Retired officer, Federal Reserve Bank	9
M. A. Brokaw	Owner, Penny Whistle Toys	2
	Total directorships	17
	Total directorships/outside directors	17/5 = 3.4
	Percentage with three or more directorships	40%
	Is the board busy? $(0 = No, 1 = Yes)$	No
	Panel C: Clorox—Outside Directors 1993	
D. Boggan	Vice Chancellor, U.C. Berkeley	1
D. O. Morton	Retired COO, Hewlett Packard	5
E. L. Scarff	Former CEO, Arcata Corporation	1
L. R. Scott	CEO, Carolina Freight	3
F. N. Shumway	Retired Chairman, Allied Signal	4
J. A. Vohs	Retired Chairman, Kaiser Health GP	2
	Total directorships	16
	Total directorships/outside directors	16/6 = 2.66
	Percentage with three or more directorships	50%
	Is the board busy? $(0 = \text{No}, 1 = \text{Yes})$	Yes

(continued)

Table III—Continued

Director		Main Occ	Total Directorships		
	Pa	anel D: MGM Grand–	-Outside Directors 1993		
Willie D. Davis	Presid	ent, All-Pro Broadcas	ting	8	
Lee A. Iacocca		nan, Iacocca Capital (ysler)	GP (Retired CEO,	1	
E. Parry	CEO,	Valley Capital Corpor	ation	2	
•	Total o	directorships		11	
	Total directorships/outside directors				
	Percen	tage with three or mo	33.33%		
	Is the	No			
Boards with Outs	ide				
Directors Holding	Three	Are Outside	Mean Directorships	Mean Directorships	
or More Directors	hips	Directors Busy?	per Outside Director	per Board	
		Pan	el E		
$x \ge 75\%$		Yes	3.35	1.85	
$50\% \le x < 75\%$		Yes	3.19	1.77	
25% < x < 50%		No 3.41		1.88	
$x \le 25\%$		No	2.36	1.38	

this measure is unlikely to reflect the value of future investment opportunities. In addition, we suspect that financial constraints are less likely to be predominant in our sample, which consists of the largest U.S. corporations during the time period studied.

A. Market-to-Book Ratio Tests

We estimate firm-fixed effects regressions using the market-to-book ratio as the dependent variable. We calculate the market-to-book ratio as the market value of the firm's equity at the end of the year plus the difference between the book value of the firm's assets and the book value of the firm's equity at the end of the year, divided by the book value of the firm's assets at the end of the year. This calculation closely follows that of Smith and Watts (1992). The regressions control for corporate governance and financial characteristics likely to affect firm performance. Gilson (1990) finds that during periods of financial distress, firms reduce board size, and Yermack (1996) documents a negative and significant association between company valuation and board size. We include the log of board size in our tests. We control for firm size using the natural log of sales. Board composition is controlled for by scaling the number of outside directors by board size. We include the percentage of the firm's common shares beneficially owned by company insiders as an independent variable, because several studies link share ownership with firm value. We also include the natural log of meetings and the number of board committees as independent variables (Vafeas (1999)). We control for both the presence of interlocking directorships between outside directors and the CEO using an indicator variable, and for the number of outside directorships held by the CEO (Booth and Deli (1996)). Our regressions include the ratio of depreciation expenditures to sales as a measure of the firm's investment opportunity set (tests using alternative measures are described later), and also control for firm age. Throughout, the fixed effects specification is employed to control for unobservable attributes, such as company's history, culture, and product mix, that potentially affect firm performance.

The results of the multivariate models are reported in Table IV. Model (1) shows that the coefficient for the busy board indicator is negative and statistically significant at the 1% level. In model (2), we use the percentage of outside directors that are busy and find a negative and significant coefficient on this variable as well. Therefore, both specifications indicate a negative and statistically significant relation between the presence of busy outside directors and the market-to-book ratio. Our estimates suggest that the impact of busy outside directors on firm performance is economically nontrivial. The coefficient estimate in model (1) indicates that a busy board reduces the market-to-book ratio by about 0.04.

We examine if the marginal impact of a busy outside director depends on whether or not a majority of the outside directors are busy. Model (3) includes an interaction term between the percentage of busy outside directors and the busy board indicator variable. The interaction term is negative and significant at the 6% level, indicating that when a majority of outside directors are busy, the market-to-book ratio has a stronger negative association with the percentage of busy outside directors. This suggests that reducing the fraction of busy directors for boards in which a majority of outside directors are busy is likely to yield more meaningful valuation improvements.

Coefficient estimates for the control variables are in line with those reported by other studies. We obtain an inverse and statistically significant association between board size and firm performance (Yermack (1996)). The number of business segments is negatively related to performance (Berger and Ofek (1995)), while ownership by officers and directors yields positive coefficients (Yermack (1996)). As in Fich and Shivdasani (2005), we find that firm size is positively associated with market-to-book ratio. Market-to-book ratios are also negatively related to firm age and the presence of an interlocking board, though the latter effect is significant at the 10% level in some specifications.

Using the fixed effects framework, we are able to replicate the cross-sectional results of Ferris et al. (2003) in our sample. Ferris et al. (2003) measure the degree to which directors are busy by using the average numbers of directorships per director and directorships per outside director. Models (4) and (5) show that neither of these two variables displays a significant association with the market-to-book ratio; similar to the results obtained by Ferris et al. (2003). The contrast between these results and those shown in models (1)–(3) suggest that inferences on the effects of busy boards are sensitive to how the presence of busy directors is measured.

Table IV Busy Outside Directors and Firm Performance

This table presents fixed effects regressions of firm performance and busy outside directors. All regressions use market-to-book ratio as the dependent variable. We calculate the market-to-book ratio as the market value of the firm's equity at the end of the year plus the difference between the book value of the firm's assets and the book value of the firm's equity at the end of the year, divided by the book value of the firm's assets at the end of the year. This calculation closely follows that of Smith and Watts (1992). Regression (1) uses a (0, 1) dummy variable equal to one if 50% or more of the board's outside directors individually hold three or more directorships as the key independent variable. Regression (2) uses the percentage of outside directors that hold three or more directorships (i.e., are busy) as the key independent variable. We classify boards as being interlocked if the CEO sits on the board of an outside director; all other variables are self-explanatory or are described in the main text. The sample is described in Panel A of Table II. We report White (1980) heteroskedasticity-robust p-values in parentheses below each coefficient estimate.

Variable	(1)	(2)	(3)	(4)	(5)
Board Characteristics					
Average directorships by outside directors				-0.077	
				(0.26)	0.040
Average directorships by board					-0.040 (0.60)
Busy outside directors (0, 1)	-0.042				(0.60)
Busy outside directors (0, 1)	(0.00)				
Percentage of busy outside directors	(0.00)	-0.152	-0.083		
		(0.00)	(0.00)		
Percentage of busy outside directors × Busy			-0.071		
outside directors (0, 1)					
			(0.06)		
Log of the directorships held by the CEO	-0.166	-0.169	-0.160	-0.179	-0.177
	(0.16)	(0.13)	(0.13)	(0.09)	(0.12)
Firm has an industry director	0.050	0.049	0.044	0.048	0.049
	(0.28)	(0.54)	(0.60)	(0.32)	(0.23)
Directors' ownership (% of common)	0.187	0.122	0.124	0.188	0.188
	(0.09)	(0.08)	(0.07)	(0.10)	(0.08)
Board interlock (0, 1)	-0.009	-0.008	-0.008	-0.014	-0.010
	(0.07)	(0.07)	(0.08)	(0.05)	(0.05)
CEO ownership (% of common)	0.008	0.015	0.016	0.009	0.009
	(0.12)	(0.08)	(0.09)	(0.13)	(0.13)
Log of board size	-0.314	-0.290	-0.298	-0.303	-0.299
_	(0.01)	(0.05)	(0.05)	(0.01)	(0.05)
Log of board meetings	-0.091	-0.119	-0.100	-0.093	-0.090
	(0.26)	(0.40)	(0.27)	(0.22)	(0.29)
Board committees	-0.016	-0.013	-0.009	-0.011	-0.015
	(0.68)	(0.58)	(0.47)	(0.56)	(0.64)
Board composition (% outside directors)	0.165	0.147	0.149	0.161	0.161
•	(0.06)	(0.24)	(0.20)	(0.06)	(0.06)
Firm Characteristics					
Return on assets	2.002	2.044	2.029	1.996	2.004
	(0.00)	(0.01)	(0.00)	(0.00)	(0.00)
Firm size (log of total sales)	0.433	0.436	0.441	0.430	0.438
	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Firm age	-0.001	-0.001	-0.001	-0.001	-0.001
	(0.01)	(0.01)	(0.01)	(0.01)	(0.01)
Growth opportunities	0.077	0.093	0.080	0.100	0.079
(depreciation expense/sales)					
	(0.24)	(0.27)	(0.25)	(0.31)	(0.26)
Number of business segments	-0.049	-0.051	-0.048	-0.052	-0.049
ŭ	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
Year (0, 1) indicators	Yes	Yes	Yes	Yes	Yes
Adjusted R^2	37.53%	37.69%	38.11%	33.02%	34.18%

B. Operating Performance Tests

The market-to-book ratio is also often used as a measure of growth opportunities. Despite our controls for investment opportunities in the regressions, and additional robustness tests described in Section IV, we are concerned about the possible impact that growth opportunities have on our coefficient estimates. To address this issue, we estimate the impact of busy boards on accounting measures of current performance, since these measures are less likely to be mechanically driven by growth opportunities. The fixed effects regressions in Table V replace the market-to-book ratio with three different measures of operating performance.

Models (1) and (2) of Table V use the ROA as the dependent variable.⁶ These regressions produce results that are consistent with those in Table IV. For example, in model (1), the coefficient for the busy board indicator variable is negative and statistically significant (-0.0024, p-value = 0.00). This estimate indicates that ROA is about 0.24 percentage points lower in firms with busy boards. Therefore, while the effect of a busy board on ROA is statistically significant, the economic magnitude of the relation is not particularly large.

We also measure firm performance using two additional financial ratios, namely, sales over assets (asset turnover ratio), and the return on sales, computed as operating income over net sales. We estimate fixed effects regressions using these ratios as dependent variables, and present them as models (3) and (4) of Table V. The busy board indicator yields a negative and significant coefficient of -0.033 with a p-value =0.02 in the sales over assets regression, and a -0.0027 coefficient with a p-value =0.00 in the return on sales regression. These results are consistent with our earlier findings and suggest that companies with busy boards tend to display weaker operating profitability than firms in which boards are not busy.

C. Robustness Checks

C.1. Alternative Hypothesis

While the preceding results support the view that busy outside directors are associated with lower firm performance, the findings could be consistent with other explanations. Gilson (1990) reports that distressed firms revamp their boards by making them more independent and by appointing turnaround specialists. It is possible that busy outside directors tend to be appointed to boards of poorly performing companies if these directors are viewed as helpful in formulating turnaround strategies. To control for this potential endogeneity, we reestimate our regressions using 1- and 2-year lagged values of the busy board

⁶ We calculate ROA as operating income before depreciation (Compustat item 13) plus the decrease in receivables (Compustat item 2), the decrease in inventory (Compustat item 3), the increase in current liabilities (Compustat item 72), and the decrease in other current assets (Compustat item 68). We scale this measure by the average of beginning- and ending-year book value of total assets (Compustat item 6).

Table V Fixed Effects Coefficient Estimates: Busy Outside Directors and Firm Profitability

In this table, the dependent variables are return on assets (ROA), sales over assets, and return on sales. We first sum operating income before depreciation (Compustat item 13) plus the decrease in receivables (Compustat item 2), the decrease in inventory (Compustat item 3), the increase in current liabilities (Compustat item 72), and the decrease in other current assets (Compustat item 68). We scale this measure by the average of beginning- and ending-year book value of total assets (Compustat item 6) to find ROA. Similarly, we divide this measure by the average of beginning- and ending-year sales to compute ROS. We use the log of total capital as a proxy for firm size. Regressions (1), (2), and (3) use a (0,1) dummy variable that equals one if 50% or more of the board's outside directors individually hold three or more directorships as the key independent variable. Regression (2) uses the percentage of outside directors that hold three or more directorships (i.e., are busy) as the key independent variable. All other variables are self-explanatory or are described in the main text. The sample consists of *Forbes* 500 firms from 1989 to 1995 described in Panel A of Table II. White (1980) heteroskedasticity-robust p-values appear in parentheses below each coefficient estimate.

		Depender	nt Variable	
Independent Variables	(1) ROA	(2) ROA	(3) Sales/Assets	(4) ROS
Board Characteristics				
Busy outside directors (0, 1)	-0.00235		-0.033	-0.00272
	(0.00)		(0.02)	(0.00)
Percentage of busy outside directors		-0.0163		
		(0.01)		
Log of the directorships held by the CEO	-0.078	-0.071	-0.002	-0.041
	(0.27)	(0.20)	(0.61)	(0.33)
Firm has an industry director	0.020	0.015	0.004	0.018
	(0.31)	(0.33)	(0.40)	(0.39)
Directors' ownership (% of common)	0.022	0.025	0.222	0.024
	(0.17)	(0.11)	(0.09)	(0.11)
Board interlock (0, 1)	-0.005	-0.005	-0.004	-0.005
	(0.10)	(0.13)	(0.06)	(0.08)
CEO ownership (% of common)	0.003	0.003	0.141	0.005
	(0.17)	(0.29)	(0.29)	(0.13)
Log of board size	-0.041	-0.043	-0.139	-0.032
	(0.01)	(0.01)	(0.04)	(0.01)
Log of board meetings	-0.129	-0.134	-0.099	-0.138
	(0.05)	(0.06)	(0.11)	(0.02)
Board committees	-0.000	-0.000	-0.005	-0.000
	(0.40)	(0.42)	(0.44)	(0.39)
Board composition (% outside directors)	0.002	0.002	0.003	0.007
•	(0.35)	(0.38)	(0.45)	(0.11)
Firm Characteristics				
Return on sales (1) and (2),	1.841	1.967	3.671	4.698
Return on capital (3) and (4)	(0.00)	(0.00)	(0.00)	(0.00)
Firm size	0.048	0.047	0.166	0.094
	(0.00)	(0.00)	(0.01)	(0.03)
Firm age	-0.0008	-0.0008	-0.0006	-0.0008
	(0.03)	(0.04)	(0.14)	(0.03)
Depreciation expense/sales	0.054	0.050	0.063	0.060
	(0.06)	(0.05)	(0.07)	(0.13)
Number of business segments	-0.006	-0.006	-0.003	-0.008
S	(0.05)	(0.04)	(0.00)	(0.03)
Year (0, 1) indicators	Yes	Yes	Yes	Yes
Adjusted R^2	26.36%	27.10%	13.90%	25.01%

indicator and other corporate governance variables. These tests continue to yield an inverse and statistically significant association between firm performance and our busy board measures. We describe more detailed tests of this potential endogeneity in Section IV.

C.2. Size and Performance Proxies

We repeat the analyses presented in Table IV using different proxies for firm size, replacing the natural log of sales by both the natural log of capital and the natural log of assets. These tests also yield an inverse association between busy board and performance. Our result continues to be robust to different constructions of the dependent variable. Instead of the Smith and Watts (1992) market-to-book ratio calculation, we use the Tobin's Q calculation of Perfect and Wiles (1994), and the Q calculation of Shin and Stulz (2000). These different constructions of the dependent variable do not qualitatively alter the results.

C.3. Characterizing Busy Outsiders

We use a less expansive definition of our key independent variable based on a slightly different procedure to identify busy outside directors. Core et al. (1999) differentiate between outside directors that are currently employed and those that are retired. In their taxonomy, retired outside directors are considered busy if they serve on six or more publicly traded boards. We follow their definition and deem employed outside directors busy when they hold three or more directorships and retired outside directors busy when they hold six or more directorships. A board is defined as busy when 50% or more of its outside directors are individually classified as busy. We construct a (0, 1) indicator under this criterion and perform regressions similar to those in Table IV. The coefficient estimate for a (0, 1) independent variable under this taxonomy is -0.0401 (p-value =0.06). This estimate is slightly smaller in magnitude than that reported in Table IV, but generates qualitatively similar inferences.

C.4. Investment Opportunities

Notwithstanding the results in Table V, a concern with the regressions presented in Table IV is whether we appropriately control for the role of the firm's investment opportunity set. As an alternative to using depreciation to control for investment opportunities, we use the ratio of capital expenditures to sales and obtain results similar to those reported earlier. We recognize the possibility

⁷ Total capital adds the market value of the firm's equity, book value, long-term debt, and an estimated market value of preferred stock. We calculate the market value of preferred stock by dividing preferred dividends over the prevailing yield on Moody's index of high-grade industrial preferred stocks.

that in the presence of financial constraints, growth opportunities may not be fully captured by capital expenditures. Therefore, we also use the ratio of research and development (R&D) to sales, the earnings-to-price ratio, and the variance of common stock returns as other control variables. The use of these different proxies for investment opportunities does not alter our results. Our proxies for busy outside directors continue to yield a negative and significant association with the market-to-book ratio in all specifications.

D. Summary

Results presented in this section indicate that firm performance, measured using both the market-to-book ratio as well as several measures of operating profitability, is inversely related to the presence of a majority of outside directors that serve on three or more boards. However, similar to Ferris et al. (2003), we are unable to uncover such a relation using the average number of board seats held by all directors or by outside directors.

Our estimates suggest that a change in the board's status from busy to non-busy is associated with an increase in the market-to-book ratio of 0.04. To put this result in perspective, findings in Gompers, Ishii, and Metrick (2003) imply that during the 1990–1995 period, a one point increase in their composite "Governance Index" reduces the market-to-book ratio by an average 3.37 percentage points. Yermack (1996) suggests that an increase in board size from eight to nine directors leads to a reduction in market-to-book ratio of 0.04 and Daines (2001) finds that incorporation in Delaware leads to a 0.06 increase in market-to-book ratio. Anderson and Reeb (2003) find market-to-book ratio is about 0.15 higher for family-run firms and Fich and Shivdasani (2006) find it is about 0.14 higher for firms with stock option plans for outside directors.

IV. Appointments and Departures of Busy Outside Directors

Results of Section III show a negative association between busy boards and firm performance. In this section, we turn to the potential endogeneity of busy outside directors with respect to performance. We explore whether firms tend to appoint busy directors when performance suffers and/or whether non-busy directors are more likely to depart the board when firms perform well. We therefore conduct tests on the number of board seats held by directors and on appointments of new outside directors. We also examine the determinants of outside director departures. Our primary focus in these tests is whether patterns in appointments and departures of outside directors explain the negative relation between firm performance and busy boards described in Section III.

There are several reasons why appointments of directors with multiple board seats might be linked to company performance. It is possible that poorly performing firms are more likely to seek out new outside directors that sit on several boards because such directors have valuable reputations and experience

that can help reverse poor performance. An alternative possibility is that poorly performing firms may find it difficult to attract directors that have high reputations and significant opportunities to serve on other boards.⁸

Similarly, reputational concerns may also affect how firm performance influences the departure of outside directors. Brown and Maloney (1999) suggest that directors with significant reputational capital might choose to protect it by leaving boards of companies that perform poorly. Alternatively, if poor firm performance causes CEOs to favor busy directors that might be weaker monitors, they may choose to reappoint outside directors with multiple board seats, while denying reappointment to those serving on few boards. To understand how firm performance affects changes in board composition, we study board appointments and departures within our sample.

As described in Section II, our sample firms appointed 2,314 outside directors during 1989 to 1995. We track each of these outside directors until the year 2000 to determine which of these directors remained on the board and which subsequently departed the board. For each outside director appointed, we review both the annual report and the firm's proxy statements to establish whether the appointed director remained on the board. We search the Wall Street Journal Index and Lexis/Nexis when we are able to identify a departure, and read newspaper stories and company press releases in order to ascertain the reason for the departure. We identify a total of 1,676 director departures among our sample. Of these, we are able to identify 360 voluntary departures. We classify a departure as voluntary if the reason given for the director's departure is either to pursue other interests or to take a position elsewhere. We also record 490 departures related to board term limits, normal retirements, health problems, or death. In 826 instances, we are unable to precisely establish the reason for the departure. Of the 2,314 appointees, 638 continued serving as directors until the end of year 2000.

We conduct four tests using this sample of outside director appointments and departures. First, we estimate a maximum likelihood model of the number of board seats held by appointees. Second, we examine the factors that affect the likelihood that a busy director is appointed to the board. Third, we estimate a hazard model to understand the determinants of outside director departures. Our fourth test examines the probability that a busy director departs the board. In all tests, our primary focus is to understand whether firm performance has a significant impact on the types of outside directors that join and leave the board.

A. Multivariate Analysis of the Determinants of Directorships

We estimate a Poisson maximum likelihood regression to investigate the determinants of directorships for the 2,314 appointees. The dependent variable is the count of the directorships held by each outside director. We include

⁸ This potential endogeneity, however, works against uncovering the negative relation that we document between firm performance and busy outside directors.

the industry-adjusted stock return over the prior year as a measure of the appointing firm's performance as an independent variable. The regression includes appointee-specific characteristics such as age, gender, and educational and professional qualifications. We also include firm-specific attributes relating to the companies in which the individual serves as a director. Unless the appointee is the CEO of another firm, we compute the average stock ownership by the outside director for all of the boards on which he/she serves, as well as the average industry-adjusted ROA, the market-adjusted stock return, and the average size (natural log of sales) of these firms. If the appointee is a CEO in another firm, we record the stock ownership, industry-adjusted ROA, the market-adjusted stock return, and the size of the firm in which he/she serves as CEO.

The results of the Poisson model are reported in the first column of Table VI.⁹ We find that the performance of the appointing firm is unrelated to the count of directorships held by outside director appointees. In contrast, the average performance of the firms on whose boards the directors sit is positively associated with their directorship count. The coefficients on both the average directorship industry-adjusted ROA and the market-adjusted stock return are positive, with *p*-values of 0.03 and 0.07, respectively.

We also observe that being a current or retired CEO of another firm positively affects the number of directorships held as does being a director at larger companies. Similar results are documented in Fich (2005) and Ferris et al. (2003) and suggest that the increased visibility from sitting on boards of large companies may help some directors obtain more directorships. Finally, we find a lower count of directorships when directors have gray status at other boards, suggesting that firms avoid appointing board members that face potential conflicts of interest at other companies. Alternatively, extensive business dealings with a firm may leave gray directors with little time to serve on other boards.

Overall, the results of the Poisson model indicate that the accumulation of directorships is positively related to the performance of the firms in which the individual is an outside director, but we do not find evidence that poor performance increases the frequency of appointments of outside directors that serve on several boards.

⁹ The Poisson model specifies that if λ is defined by $\log(\lambda) = \mathbf{X}\beta$, where \mathbf{X} is a vector of independent variables and β is a parameter vector, then the probability of n outside directors obtaining a directorship in a given year is given by: $\lambda^n e^{-\lambda}/\lambda!$ The log-likelihood function of this specification is maximized over β to produce maximum likelihood estimates and is given as,

$$L(\beta) = \sum_{i=1}^{N} \sum_{t=1}^{T} \left\{ C_1 - \mathrm{e}^{(\mathbf{X}_{\mathrm{it}}\beta)} + n_{it}\mathbf{X}_{it}\beta \right\},$$

where C_1 is a constant that does not change the maximization process, N is the number of firms, T is the number of time periods per firm, and n_{it} is the number of outside directors obtaining a directorship in firm i in year t.

B. Appointments of Busy Outside Directors

To study appointments of busy outside directors, we estimate a logit model using the 2,314 appointees, where the dependent variable is set equal to one if the director holds three or more total directorships (i.e., is busy), and is set

Table VI Determinants of Directorships and Appointments of Busy Outside Directors

Model (1) presents Poisson maximum likelihood estimates for the determinants of the number of directorships held by outside directors. The dependent variable counts the number of directorships held by the outside director. Model (2) presents logit estimates for busy directors. The dependent variable takes the value of one if the outside director holds three or more total directorships and the value of zero otherwise. The sample consists of 2,314 outside directors appointed to the boards of our 508 sample firms from 1989 to 1995. Unless the director is a CEO of another firm, we compute the average ownership of the outside director on all of the boards he serves, as well as the average industry-adjusted ROA and the size of these firms. If the appointee is a CEO in another firm, we simply record his ownership, the industry-adjusted ROA, and the size of the firm in which he is the CEO. We use the natural log of sales to proxy for firm size in model (1) and the natural log of the market value of assets in model (2). All industry adjustments are done by subtracting the median of the variable matching by the company's two-digit SIC code. We report p-values under parentheses.

Variable	(1) Poisson	(2) Logit
Constant	-0.197	-2.818
	(0.42)	(0.00)
Appointing Firm's Performance		
Industry-adjusted stock return $(R_t - R_{ind})_{t-1}$	0.108	0.152
	(0.36)	(0.34)
Sales growth $[\log(Sales_t/Sales_{t-1})]$	-0.405	-0.121
	(0.49)	(0.53)
Appointee's Characteristics		
Age	-0.057	-0.166
	(0.01)	(0.28)
Gender (Female $= 1$, Male $= 0$)	0.409	0.108
	(0.20)	(0.12)
Average directorship ownership (% of common stock)	0.006	-0.105
	(0.59)	(0.01)
Average directorship industry-adjusted ROA	0.190	0.377
	(0.03)	(0.00)
Average directorship change in the stock return $(R_t - R_{mkt})$	0.095	0.120
	(0.07)	(0.01)
Average directorship firm size	0.167	0.219
•	(0.00)	(0.01)
CEO in another firm	0.202	0.883
	(0.00)	(0.00)
Retired CEO in another firm	0.288	1.659
	(0.00)	(0.00)
Gray director in another firm	-0.040	-0.199
	(0.05)	(0.01)
Law degree	-0.240	0.310
-	(0.50)	(0.11)

(continued)

Table VI—Continued

Variable	(1) Poisson	(2) Logit
MBA	0.194	-0.004
	(0.25)	(0.55)
Academic	-0.521	-0.184
	(0.28)	(0.05)
Politician (active or retired)	0.111	0.969
	(0.12)	(0.05)
Appointing Board's Characteristics		
Independent board	-0.329	-0.633
-	(0.10)	(0.01)
Board size	0.189	0.327
	(0.04)	(0.02)
New CEO appointed	-0.041	-0.193
••	(0.52)	(0.10)
Founder	-0.177	-1.041
	(0.20)	(0.28)
Pseudo R^2	0.03	0.11
N	2,314	2,314

equal to zero otherwise. Independent variables are similar to those used in the Poisson estimation.

Model (2) of Table VI shows that firm performance for the appointing firm is not significantly associated with the probability of appointing a busy outside director. This result casts doubt on the idea that poor firm performance makes the appointment of a busy outside director more likely. As with the Poisson model estimates, we find that appointments of busy outside directors are more likely when the other firms in which they hold directorships perform better. The coefficients are positive and statistically significant for variables that measure the average industry-adjusted ROA and the average market-adjusted stock return for companies in which the appointees are incumbent directors. In addition, busy outside directors are more likely to be appointed if they tend to sit on boards of larger firms, and are current or retired CEOs of other companies. Appointment of a busy director is less likely if the director is a gray director on another board or if the director is in academia. However, the probability of a busy director appointment is higher if the director is an active or retired politician.

Finally, we find that appointments of busy outside directors are less likely if these directors have relatively high stock ownership on the other boards on which they sit. One interpretation of this result is suggested by Perry and Peyer (2005), who argue that executives with high stock ownership at their own firms will be reluctant to assume outside board appointments if doing so detracts from maximizing value at their own companies. Our result suggests a similar interpretation—outside directors with strong ownership incentives on their current boards might be reluctant to take on additional board seats since that constrains their monitoring efforts on their current boards.

C. Departure Hazard for Outside Directors

We use a hazard model to investigate the annual risk of departure for each director. The hazard specification is appropriate because it censors all departures related to board term limits or death as well as the observations for directors that continue serving at the end of our coverage period.¹⁰

The results of the hazard model are presented in model (1) of Table VII. We find that poor firm performance is associated with a higher likelihood of outside director departure. The coefficient estimate on industry-adjusted stock return is negative and significant at the 1% level, similar to the results in Hermalin and Weisbach (1988). The results also reveal that the number of directorships is significantly related to the chance of departure. The coefficient for the number of directorships is positive and significant (0.27, p-value = 0.00) indicating that each additional directorship leads to a significant increase in the hazard of voluntary departure. In addition, departures are about 14.4% more likely to occur in firms that appoint a new CEO during the year. However, directors that are CEOs of other companies are less likely to depart the board.

D. Firm Performance and Busy Outside Director Turnover

Though the results of the hazard model indicate that directors with multiple board seats are more likely to depart, it is unclear whether firm performance affects the probability of departure of busy outside directors. To investigate this issue, we exclude departures related to board term limits or death and divide the remaining 1,186 departures by whether or not the departing director holds three or more directorships. We estimate a logit model in column (2) of Table VII, where the dependent variable takes the value of one if the departing outside director holds three or more directorships (i.e., is busy) and takes the value of zero otherwise.

The results illustrate a clear link between company performance and the departure of busy outside directors. The coefficient estimate for industry-adjusted stock return is negative and statistically significant (*p*-value of 0.05). In terms of the marginal effects implied by the coefficient on stock returns, underperforming the industry by 50% increases the probability of busy outside director departure by 1.54 percentage points. Though this effect is not large, it is meaningful when compared to the unconditional frequency of outside director turnover of 6.7% in the sample.

¹⁰ The statistical significance of a covariate is given by the log-likelihood ratio statistic. Under the null hypothesis that $\beta=\beta_0, -2\log[\frac{L(\beta_0)}{L(\beta^{\rm MLS})}]$ is distributed χ^2 with K degrees of freedom, where K is the number of elements of β , and $L(\beta)/L(\beta^{\rm MLS})$ is the likelihood ratio statistic. For the hazard estimation, presented as model (1) of Table VII, we estimate p-values for the test that an individual covariate is zero. Below each covariate, we report its risk ratio in parentheses, a transformation of the estimated coefficient that is easier to interpret. The risk ratio is defined as the ratio of the hazard function under one set of covariates (X') to the hazard under a base case set of covariates (X^0) : $R(X',X^0) \equiv \frac{h_L(t|X^0)}{h_L(t|X_0)}$. For example, the percentage change in the hazard given a one-unit change in the kth quantitative covariate is simply estimated as $\frac{h_L(t|X_0-X^0)}{h_L(t|X_0-X^0)} - 1 = \exp[\beta_k(\hat{X}+1) - \beta_k(\hat{X})] - 1 = \exp[\beta_k] - 1$.

Table VII Outside Director Departure

Estimates for the departure of outside directors appointed to the boards of Forbes 500 firms. We follow directors appointed between 1989 and 1995 until the year 2000. We are able to identify a total of 360 voluntary departures during our coverage period. We classify a departure as voluntary if the reason given for the director's departure is either to pursue other interests or to take a position elsewhere. Model (1) estimates a departure hazard regression, where risk ratios appear in parentheses below each covariate estimate. Model (2) estimates a logit regression, where the dependent variable equals one if the departing director holds three or more directorships, and equals zero otherwise. In 826 instances, we are unable to exactly establish the reason for the departure. From both models, we exclude observations related to 638 executives that continue serving as directors at the end of our coverage period. Excluded from the logit estimation are 490 departures related to board term limits, normal retirements, health problems, or death. We report risk ratios below each covariate estimate in the hazard model and heteroskedasticity-robust p-values below each logit estimate. In both models, *, **, and *** denote significance at the 1%, 5%, and 10% levels, respectively.

Variable	(1) Hazard (All Departures)	(2) Logit (Busy versus Nonbusy)
Constant		-7.200
		(0.00)
Departure Firm's Performance		
Industry-adjusted stock return $(R_t - R_{ind})_{t-1}$	-0.882^{*}	-0.579^{**}
	(0.414)	(0.05)
Sales growth $[\log(Sales_t/Sales_{t-1})]$	-0.130	-0.089
	(0.878)	(0.39)
Departing Director Characteristics		
Directorships	0.271^{*}	
	(1.312)	
Age (years above 55 in the hazard)	0.078***	0.085
	(1.081)	(0.14)
Gender (Female $= 1$, Male $= 0$)	0.213	0.651**
	(1.238)	(0.04)
Ownership (% of common stock)	-0.398	0.332
•	(0.671)	(0.64)
CEO of another firm	-0.526**	0.153
	(0.591)	(0.26)
Retired CEO of another firm	-0.243	-0.056
nemed one of anomer min	(0.784)	(0.45)
Departure Board Characteristics	(0.101)	(0.13)
Independent board	-0.108	0.128*
macpondoni bodi d	(0.898)	(0.01)
Board size	0.019	-0.251^{**}
Board Size	(1.019)	(0.05)
New CEO appointed	0.135*	1.055*
New CEO appointed	(1.144)	(0.00)
Founder	-0.515	-0.611
rounder		-0.611 (0.15)
	(0.597)	
N	1,676	1,186
Year (0, 1) indicators	No	Yes
Model's p -value	0.00	0.00

In addition to performance measures, we find that aspects of the company's governance structure impact the probability of busy outside director departures. Busy directors are more likely to leave small and/or independent boards. These results suggest that their departures are more likely when the board is predisposed to stronger monitoring.

We also find that busy outside directors are more likely to depart when a new CEO takes office. In terms of the marginal effects implied by this coefficient estimate, if a new CEO is appointed, the probability of a simultaneous busy director departure increases by 4.78 percentage points. Hermalin and Weisbach (1988), Farrell and Whidbee (2000), and Yermack (2004) find that outside directors are more likely to leave the board when a new CEO is appointed. Our results illustrate that this effect is more pronounced for busy outside directors. To put our results in perspective, unconditionally, a new CEO increases the probability of an outside director departure in the same year by 2.98 percentage points. Therefore, our estimates suggest that busy outside directors are substantially more likely to leave the board than nonbusy outsiders when new CEOs are appointed.

In summary, we uncover evidence suggesting that the presence of busy outside directors is endogenous with respect to firm performance. However, since busy directors are more likely to depart following poor firm performance, the effect that we find works against our documented results on the inverse relation between performance and a majority of busy outside directors.

V. Busy Boards and Forced CEO Turnover

As an additional test of the impact of busy boards on the quality of corporate governance, we study the sensitivity of CEO turnover to firm performance in our sample. Besides providing direct evidence on how busy directors impact an important board decision, the advantage of this approach is that it is immune to the potentially confounding effect of growth opportunities that might be present in the market-to-book ratio regressions presented earlier.

We estimate logit models for the probability of forced CEO turnover while controlling for several firm and corporate governance attributes. We classify a turnover episode to be forced using a three-part rule similar to that in Parrino (1997). First, we classify turnover as forced if the *Wall Street Journal* reports that the CEO is fired, forced from office, or departs due to policy differences. Second, we classify turnover to be forced if the CEO is not close to retirement (60 years or more) and the *Wall Street Journal* does not report the reason for the departure as being death, poor health, or the assumption of another position (elsewhere or within the firm). Third, we classify a turnover episode as forced if the CEO is not of retirement age and the *Wall Street Journal* reports that the CEO is retiring, but does not announce the retirement at least 6 months prior to the departure.

CEO turnover episodes occur for 11.28% (321 events) of the observations in our sample, a frequency similar to those in other studies such as Coughlan and Schmidt (1985) who report a 12% frequency using 1978 to 1980 data, and

Mehran and Yermack (1997) who report a 10.8% frequency using 1984 to 1991 data. Approximately, 63% of departing CEOs (202 executives) leave office due to retirement, and 18% (58 executives) are forced out of office, an incidence rate comparable to that in Parrino, Sias, and Starks (2003) who document a 19% frequency using 1982 to 1993 data.

Our independent variable of interest is the busy board (0, 1) indicator that takes the value of one if 50% or more outside directors hold three or more directorships. As a measure of performance, we use the industry-adjusted ROA for the year preceding the turnover. We include a (0, 1) variable to indicate an independent board, where a value of one denotes that 50% or more of the directors are outside directors. Other control variables include company size, measured as the natural log of sales, the natural log of board size, and the percentage of shares owned by institutional investors, outside directors, and the CEO. The models control for the natural log of the years the CEO has been in office. Finally, Mehran and Yermack (1997) find an inverse association between the probability of voluntary CEO turnover and the value of stock option compensation in relation to total pay. Thus, we add the Black and Scholes (1973) value of the options granted to the CEO scaled by total pay as an independent variable in the regression. ¹¹ All models include (0, 1) year indicators.

Table VIII presents the results of four logit models based on 2,844 CEO-year observations drawn from our 508 firms from 1989 to 1995. Below each estimate, we report heteroskedasticity-adjusted *p*-values in parentheses. Regression (1) confirms results from prior studies, namely, that forced CEO turnover is sensitive to firm performance. In terms of the marginal effects implied by the coefficient in model (1), underperforming the industry by 50% in the prior year increases the probability of forced CEO turnover by 4.36 percentage points.

Regression (2) includes the control variables discussed earlier. In general, we find these variables influence CEO turnover in a manner consistent with that reported in previous research. For example, CEOs are less likely to be forced out of office if they own large amounts of stock or belong to the founding family (Denis, Denis, and Sarin (1997)). As in Mehran and Yermack (1997), we find no significant association between forced turnover and stock option compensation for the CEO.

Regression (2) also confirms the effect of independent boards on the turnover–performance relation in our sample by including an interaction term between firm performance and the independent board indicator. The coefficient for this interaction term is negative (-4.12, p-value = 0.06), and the sum of the coefficients on industry-adjusted stock return and the interaction term (-7.07 = -2.95-4.12) is significantly different from zero at the 1% level. This finding indicates that the performance–turnover relation is stronger in the presence of independent boards, a result that is similar to that described in Weisbach (1988).

¹¹ The option value is adjusted for dividend payouts (Merton (1973)).

Regression (3) estimates the impact of the busy board indicator by including the interaction between the busy board indicator and firm performance. We find that busy boards have a significant impact on the sensitivity of forced CEO turnover to performance. The interaction term yields a positive

Table VIII Busy Outside Directors and Forced CEO Turnover

Logit coefficient estimates of forced CEO turnover. The dependent variable takes the value of one if the CEO is forced out of office and the value of zero otherwise. The independent variables are as follows. Busy outsiders is a dummy variable that equals one if 50% or more of the outside directors hold three or more directorships. Return on assets (ROA) is operating income before depreciation (Compustat item 13) plus the decrease in receivables (Compustat item 2), the decrease in inventory (Compustat item 3), the increase in current liabilities (Compustat item 72), and the decrease in other current assets (Compustat item 68). This measure is divided by the average of beginningand ending-year book value of total assets (Compustatitem 6). A board is classified as independent if 50% or more of its directors are independent, thus, we construct an indicator that equals one in these cases and equals zero otherwise. Firm size is the natural logarithm of total sales (Compustat item 12). CEO stock compensation divides the Black and Scholes (1973) value of the options granted during the year by base pay. CEO tenure measures the years the CEO has held the chief executive job. Ownership for the CEO, institutions, and outside directors is measured as a percentage of common. The sample consists of 2,844 CEO-year observations drawn from 508 Forbes 500 firms from 1989 to 1995. White (1980) heteroskedasticity-robust p-values are reported in parentheses below each coefficient estimate.

Variable	(1) Estimate	(2) Estimate	(3) Estimate	(4) Estimate
Constant	-3.81	-3.87	-3.88	-3.88
	(0.00)	(0.00)	(0.00)	(0.00)
Busy outsiders (0, 1) indicator			-0.23	-0.13
			(0.01)	(0.13)
Industry-adjusted ROA_{t-1}	-3.60	-2.95	-2.90	-2.88
	(0.00)	(0.00)	(0.00)	(0.00)
Independent board indicator		0.60	0.64	0.58
		(0.23)	(0.06)	(0.30)
Firm size		0.23	0.19	0.22
		(0.00)	(0.00)	(0.00)
CEO stock compensation		-0.15	-0.15	-0.17
		(0.48)	(0.46)	(0.40)
CEO tenure		-0.00	-0.00	-0.00
		(0.45)	(0.45)	(0.56)
CEO ownership		-2.83	-2.92	-2.86
		(0.00)	(0.00)	(0.00)
Institutional ownership		0.00	0.00	0.00
		(0.42)	(0.44)	(0.42)
CEO is member of the founding		-2.41	-2.37	-2.39
family		(0.00)	(0.00)	(0.00)
Outside director ownership		-0.30	-0.32	-0.27
		(0.06)	(0.04)	(0.01)
Total number of directors		0.00	0.00	0.00
		(0.28)	(0.23)	(0.27)

(continued)

Table VIII—Continued

Variable	(1) Estimate	(2) Estimate	(3) Estimate	(4) Estimate
Independent board indicator \times (Industry-adjusted ROA _{t-1})		-4.12 (0.06)		
Busy outsiders X indicator \times (Industry-adjusted ROA _{t-1})			2.82 (0.05)	
Busy outsiders X indicator \times Independent board indicator \times (Industry-adjusted ROA _{t-1})				1.54 (0.07)
Value of interaction term and Industry-adjusted ROA_{t-1} (p -value of F -test in parenthesis)		-7.07 (0.00)	-0.08 (0.71)	-1.34 (0.23)
$\begin{array}{l} Year \ (0, \ 1) \ indicators \\ Number \ of forced \ turnovers \\ Pr > \chi^2 \end{array}$	Yes 58 < 0.001	Yes 58 < 0.001	Yes 58 < 0.001	Yes 58 < 0.001

and statistically significant coefficient (p-value = 0.05), implying that the turnover–performance sensitivity is significantly lower when the outside directors are busy than when they are not busy. In fact, inspection of coefficient estimates on firm performance and the busy board interaction variable shows that CEO turnover is completely insensitive to firm performance when a majority of outside directors are busy (-0.08 = -2.90 + 2.82, p-value = 0.71). The estimates suggest that a 50% decline in industry-adjusted ROA increases the probability of CEO turnover by about 3.52 percentage points when outside directors are not busy, but only by 0.59 percentage points when outside directors are busy.

Regression (4) explores this result further in the context of Weisbach's (1988) findings on the importance of an independent board on turnover-performance sensitivities. We explore if the effect of independent boards on CEO turnover depends on whether a majority of the outside directors are busy by including a three-term interaction between the firm performance, the busy board indicator, and the independent board indicator. The coefficient on this variable measures whether the turnover-performance sensitivity when both the board is independent and a majority of outside directors are busy differs from turnover-performance sensitivity in other firms. The interaction term is positive and significant at the 7% level. The coefficient estimate implies that an independent board with busy outsiders weakens the turnover-performance relation (-1.34 = -2.88 + 1.54, p-value = 0.23). For a 50% decline in industryadjusted ROA, CEO turnover probability rises by 6.94 percentage points when the board is both independent and not busy, but by only 1.54 percentage points when the board is independent and busy. This result shows that busy outside directors severely ameliorate the effect of board independence on the turnoverperformance relation.

VI. Valuation Impact of Busy Outside Director Departures and Appointments

The case study evidence for Elaine Chao described in the introduction suggests that investors viewed her departure from the boards on which she served as favorable news. We use the incidence of outside director departures to study whether this pattern holds in a broader sample. If busy outside directors contribute to weaker corporate governance, we expect that their departure should be associated with a positive market reaction.

To focus on departures that are not perfectly anticipated due to retirements, term limits, etc., we study the 360 voluntary departure episodes in our sample. We drop observations coinciding with other major company announcements and those for which the exact date of departure cannot be established. To avoid confounding the results with information regarding a new board appointee, we also exclude 36 observations that announce a director's departure and the replacement simultaneously. This results in a sample of 243 departures. We use the standard event study methodology (Dodd and Warner (1983)) to compute ARs for the announcement date. Market model parameters are computed from 1 year of trading data preceding the event window. 12

Results of the event study are reported in Panel A of Table IX. We observe that positive and statistically significant ARs are associated with all voluntary departure episodes; however, the mean and median ARs (1.33% and 0.8%, respectively) related to the departure of busy outside directors are larger than the ARs (0.9% and 0.34%, respectively) related to the departure of nonbusy outside directors. These differences are statistically significant with a t-statistic of 1.99 and a Z-statistic of 3.16. These results are in agreement with those presented in Table I, and indicate that, on average, the departure of busy outside directors is welcome news to shareholders.

While the event study suggests that investors view the departure of busy outside directors favorably, it is possible that the departures trigger other changes that investors also favor. For example, the departure may change the balance of power between the CEO and the board (Hermalin and Weisbach (1998)). In some circumstances, the departure may be advantageous for the firm because the director might be able to help the firm in his/her new position. It is also possible that busy outside director departures serve as a signal that additional governance reforms are forthcoming. Further, since we focus on departures that are not accompanied by a concurrent replacement, it is plausible that the positive ARs may reflect the valuation impact of a reduction in board size (Yermack (1996)), though this explanation cannot account for the differential valuation effect associated with departures of busy and nonbusy directors.

Another explanation for these results, suggested by Shivdasani and Yermack (1999), is that some outside directors acquire conflicts of interest over time and switch status to gray directors. If this pattern is pervasive, our results might

 $^{^{12}}$ To control for possible bias on the market model parameters, we reestimate our ARs with simple net-of-market returns in place of market model returns. This estimation generates similar results to those we obtain with the market model parameters.

Table IX
Investor Reactions to Voluntary Departures of Outside Directors

Panels A and B present day-0 abnormal returns (ARs) associated with departure announcements of outside directors from the boards of $Forbes\ 500$ firms. Reported below each estimate are p-values from two-tailed tests using a t-test for means, and a Wilcoxon Z signed-rank test for medians.

				Par	nel A				
	All Departures		Director Has Two or Fewer Directorships Before Departure		Thr Dir	Director Has Three or More Directorships Before Departure		atistic Iean rences)	Z-Statistic (Median Differences)
Mean AR	0.0098 (0.06)		0.0090 (0.08)			0.0133 (0.00)	1.99		
$\operatorname{Median} AR$		0053	0.0034 (0.07)			0.0080 (0.00)			3.16
N		243	198			45			
				Par	nel B				
Is Board Busy After Departure?		N	Mean AR	Iean AR Median A		t-Statistic (Mean Differences)		Wilcoxon Z-Statistic (Median Differences)	
Yes		27	0.0073 (0.05)	0.0045					
No		18	0.0223 (0.00)	0.02148 (0.04)	-	2.02			1.98

instead capture the effect of replacing gray directors on the board. To examine this possibility, we verify whether the status of outside directors changed to gray during their tenure. We find that only 5 of the 198 nonbusy outside directors and 2 of the 45 busy outside directors in our subsample acquire conflicts of interest during their tenure warranting a change in classification to gray status. Using a different sample of director appointments, Yermack (2004) also reports that relatively few outside directors change their status to gray over their tenure. When we repeat the event study excluding these seven observations, we obtain returns similar to those tabulated.

Another possibility is that the remaining board is less likely to be busy after a busy director's departure. Of the 45 boards that lost a busy outside director, 18 firms switched status from a busy board to a nonbusy board. In these firms, the percentage of outside directors that were busy declined from above 50% to less than 50% after the departure. ¹³ We conduct an event study for the 45

¹³ We determine that boards have switched status from busy to nonbusy by focusing on the remaining outside directors on the board. It is possible, of course, that a firm may replace a departing busy director with a busy appointee, leaving the board's status unchanged. To account for this, we also track changes in board composition for a 6-month period following the director's departure and consider these changes in determining the board's busy status. Using this process, we identify 19 firms that switched to nonbusy status and obtain virtually identical results to those reported herein.

companies that lost a busy director. Panel B of Table IX summarizes the ARs for the 45 firms that lost a busy director, broken down according to whether the majority of the remaining outside directors on the board are busy or nonbusy. Mean and median ARs for firms in which a majority of outside directors are not busy after the departure are 2.2% and 2.1%, respectively. In contrast, ARs for the other 27 companies in which a majority of outside directors remain busy are 0.7% (mean) and 0.45% (median). The difference between the two subsamples is statistically significant using both the t-test and the Wilcoxon test. This suggests that investors react more favorably to busy outside director departures that have a significant impact on the extent to which outside directors as a group remain busy.

At first glance, our event study results appear to be at odds with those reported in Ferris et al. (2003). Ferris, Jagannathan, and Pritchard report a significantly positive 2-day announcement effect around the first appointment of a busy director to the board. However, this discrepancy arises because Ferris, Jagannathan, and Pritchard do not focus on firms that already have busy directors on the board. Specifically, they study 84 firms that announced the appointment of a busy director and report a positive mean announcement return of 0.41% for this sample. Among these 84 firms, 41 observations represent the first appointment of a busy director to the board, and for this subsample, the mean announcement return is 2.12%. Though not reported by Ferris, Jagannathan, and Pritchard, one can infer that the mean announcement return for the remaining 43 firms that already had at least one busy director on the board is -1.22%. Therefore, their data also suggest that director appointments that lead to several outside directors being busy have a negative effect on firm value.

We provide further evidence on the valuation impact of multiple directorships in Table X. We focus on how the stock prices of a director's existing directorship firms react when the director receives an additional board seat. For this test, we use the sample of 2,314 appointments of outside directors to our sample firms' boards. For each of these appointments, we study the announcement return for the shared-directorship firms (i.e., other firms' board on which they sit). For the shared-directorship firms, we identify 818 distinct appointment announcement dates that are uncontaminated by other major news release. Panel A shows that for the 818 shared-directorship firms, the average announcement return is -0.22% and is not significantly different from zero. However, the announcement effect depends on whether the outside director becomes busy as a result of the new board seat. In 106 cases, appointed directors held two or more board seats prior to the announcement, and are therefore classified as busy after obtaining the additional board seat. For these 106 shared-directorship firms, in which the outside director holds three or more board seats following the appointment, the average announcement return is -1.04% (p-value = 0.05). In contrast, when the director is not busy following the appointment, shared-directorship firms have a statistically insignificant average announcement return of -0.1%. The negative announcement return for shared-directorship firms when directors end up with three or more directorships suggests that the costs of the

Table X Investor Reactions to Additional Board Appointments for Outside Directors

Panels A and B present day-0 abnormal returns (ARs) for shared-directorship firms around the announcement of an additional board appointment in publicly traded companies obtained by incumbent outside directors. Reported below each estimate are p-values from two-tailed tests using a t-test for means, and a Wilcoxon Z signed-rank test for medians.

Panel A											
	All Appointments	Director Has Two or Fewer Directorships after Appointment	Director Has Three or More Directorships after Appointmen	t-Statistic (Mean t Differences)	Z-Statistic (Median Differences)						
Mean AR	-0.00224 (0.50)	-0.00102 (0.62)	-0.0104 (0.05)	2.04							
$\operatorname{Median} AR$	-0.0025 (0.44)	-0.0010 (0.31)	-0.0093 (0.04)		2.78						
N	818	712	106								
Panel B											
Shared-Dire Board Switc Busy Status Appointmen	hes to After	Mean AR	Median AR	t-Statistic (Mean Differences)	Wilcoxon Z-Statistic (Median Differences)						
Yes	23	-0.018 (0.03)	-0.0146 (0.07)								
No		-0.0083 (0.19)	-0.009 (0.24)	1.86	1.69						

additional directorship outweigh the benefits of greater reputation or experience associated with the appointment.

In Panel B of Table X, we investigate further the 106 appointments that cause outside directors of shared-directorship firms to become busy. In 23 cases, the classification of the overall board for the shared-directorship firms switches in status from nonbusy to busy as a result of the outside director's additional appointment. For these firms, the average announcement return is -1.8% and is statistically significant at the 5% level. However, the announcement return is muted in comparison and is statistically insignificant in the 83 cases for which the additional appointment does not move the board of the shared-directorship firm to busy status. Thus, the costs of an additional board seat appear to dominate the benefits for firms in which a majority of its outside directors are likely to be busy as a result of serving on multiple boards.

Overall, the results suggest that changes in busy status of the board are associated with relatively large revaluations in stock price. The departure of a busy outside director that leaves a majority of remaining outside directors as not

busy leads to an average AR of 2.2%. Evidence from the shared-directorship firms indicates an average AR of -1.8% when their boards switch to busy status as the result of an outside director gaining a board seat. The magnitude of these effects is surprisingly large, considering that the appointment of an outside director typically has a relatively muted stock price effect. For example, Rosenstein and Wyatt (1990) find that appointments of outside directors are associated with an average AR of only 0.22%. We suspect that the large valuation effects associated with busy boards reflect both the reduced monitoring of operating performance, as indicated by our results on profitability, and the lower likelihood of specific board actions, such as CEO turnover, that may have implications for firm value. During 1989 to 1994, Huson, Parrino, and Starks (2001) find that forced CEO turnover announcements led to a 4% increase in stock price, on average. We find such forced removals were less likely by busy boards. If busy boards are also distracted from monitoring managerial actions such as major acquisitions, investment proposals, and responses to takeover bids, the expected valuation effects from lax monitoring can be sizeable since these events can have a significant impact on shareholder value.

VII. Conclusions

In an era of heightened corporate governance scrutiny, substantial debate has focused on the effectiveness of the monitoring by the board of directors in large, publicly traded corporations. A significant element of this debate is centered on whether outside directors serving on several boards should cut back on the number of their directorships or be required to do so. A common view among investors and policy advocates is that serving on numerous boards can result in overstretched directors that may not be effective monitors on any board.

We present evidence confirming this popular view using a panel of large U.S. industrial firms from 1989 to 1995. We focus on boards in which a majority of outside directors sit on three or more boards, and we present several results that highlight the effect that these busy outside directors have on corporate value and governance.

We find that companies with a majority of busy outside directors display significantly lower market-to-book ratios. All else equal, firms with busy outside directors have market-to-book ratios about 4.2% lower than other firms. Evidence of weaker performance by companies with busy outside directors is also obtained from tests of operating performance. We show that firms with busy boards display lower operating ROAs, lower asset turnover ratios, as well as lower operating return on sales, and that these effects are also economically meaningful.

We show that appointments of directors who hold three or more directorships are not more likely to occur in firms that are performing poorly, casting doubt on the idea that a tendency for poorly performing firms to appoint busy directors is responsible for our findings. In fact, we find that in underperforming firms,

outside directors who hold three or more directorships are more likely to leave a board than other outside directors, suggesting that the potential endogeneity of the presence of busy directors might work against us uncovering the inverse relation between busy boards and performance.

Forced CEO turnover–performance sensitivities provide further evidence of the impact of busy outside directors on corporate governance. We find that a 50% decline in industry-adjusted performance increases the probability of turnover by 3.52 percentage points when the board is not busy, but the probability rises by only 0.59 percentage points when a majority of outside directors are busy. The joint impact of a busy board and an independent board is particularly large. Independent boards in which outsiders are not busy are associated with a 6.94 percentage point higher likelihood of CEO turnover for a 50% drop in industry-adjusted performance, but the corresponding change in CEO turnover probability is only about 1.5 percentage points higher for independent boards in which a majority of outside directors are busy.

Further support for the effects of busy outside directors comes from event study results of outside director departure announcements. We find that departures of busy outside directors are associated with significantly positive ARs that average 1.33% at the announcement. The ARs are significantly more positive when fewer than 50% of the remaining outside directors are busy after the departure.

While our tests focus on studying the costs of holding multiple board seats, findings in this paper and in the literature also point to the benefits of multiple directorships. For example, Ferris et al. (2003) show that the initial appointment of a busy director to a board without an incumbent busy director is good news for shareholders, implying that the experience or reputational impact of such outside directors can be beneficial. We also show that outside directors that are associated with well-performing firms tend to hold more board seats, consistent with the view that well-performing directors are rewarded in the market for directorships. However, results also highlight a cost to holding numerous board seats, suggesting that as directors accumulate more directorships, they may become increasingly constrained in being effective monitors, and that this effect is likely to be important when a majority of the outside directors are overcommitted.

Collectively, our results suggest that boards relying heavily on outside directors that serve on several boards are likely to experience a decline in their quality of corporate governance. However, we are reluctant to conclude with recommendations mandating limits on the number of boards on which directors should serve. Our analysis does not include other potential effects of such limits, such as the effect on directors' incentives to establish reputations as expert monitors in the market for directorships or the potential valuation effects for companies that send their executives to serve as outsiders on other boards. Nonetheless, based on the evidence in this paper, we are inclined to suggest that for some firms, the optimal board design entails a lower reliance on outside directors that serve on numerous boards.

REFERENCES

- Anderson, Ronald, and David Reeb, 2003, Founding family ownership and firm performance: Evidence from the S&P 500, *Journal of Finance* 58, 1301–1328.
- Beasley, Mark, 1996, An empirical analysis of the relation between the board of director composition and financial statement fraud, *Accounting Review* 71, 443–465.
- Berger, Philip, and Eli Ofek, 1995, Diversification's effect on firm value, *Journal of Financial Economics* 37, 39–66.
- Black, Fisher, and Myron Scholes, 1973, The pricing of options and corporate liabilities, *Journal of Political Economy* 81, 637–654.
- Booth, James, and Daniel Deli, 1996, Factors affecting the number of outside directorships held by CEOs, *Journal of Financial Economics* 40, 81–104.
- Brickley, James, James Linck, and Jeffrey Coles, 1999, What happens to CEOs after they retire? New evidence on career concerns, horizon problems, and CEO incentives, *Journal of Financial Economics* 52, 341–377.
- Brown, William, and Michael Maloney, 1999, Exit, voice and the role of corporate directors: Evidence from acquisition performance, Working paper, Claremont McKenna College.
- Coles, Jeffrey, and Chun-Keung Hoi, 2003, New evidence on the market for directors: Board membership and Pennsylvania Senate Bill 1310, *Journal of Finance* 58, 197–230.
- Core, John, Robert Holthausen, and David Larcker, 1999, Corporate governance, chief executive officer compensation, and firm performance, *Journal of Financial Economics* 51, 371–406.
- Coughlan, Anne, and Ronald Schmidt, 1985, Executive compensation, management turnover, and firm performance: An empirical investigation, *Journal of Accounting and Economics* 7, 43–66.
- Daines, Robert, 2001, Does Delaware law improve firm value? *Journal of Financial Economics* 62, 525–558.
- Denis, David, Diane Denis, and Atulya Sarin, 1997, Ownership structure and top executive turnover, *Journal of Financial Economics* 45, 193–221.
- Dodd, Peter, and Jerold Warner, 1983, On corporate governance: A study of proxy contests, *Journal of Financial Economics* 11, 401–438.
- Fama, Eugene, 1980, Agency problems and the theory of the firm, *Journal of Political Economy* 88, 288–307.
- Fama, Eugene, and Kenneth French, 1992, The cross-section of expected stock returns, *Journal of Finance* 47, 427–465.
- Fama, Eugene, and Michael Jensen, 1983, The separation of ownership and control, *Journal of Law and Economics* 26, 301–325.
- Farrell, Kathleen, and David Whidbee, 2000, The consequences of forced CEO succession for outside directors, *Journal of Business* 73, 597–628.
- Ferris, Stephen, Murali Jagannathan, and Adam Pritchard, 2003, Too busy to mind the business? Monitoring by directors with multiple board appointments, *Journal of Finance* 58, 1087–1111.
- Fich, Eliezer M., 2005, Are some outside directors better than others? Evidence from director appointments by Fortune 1000 firms, *Journal of Business* 78, 1943–1972.
- Fich, Eliezer M., and Anil Shivdasani, 2005, The impact of stock-option compensation for outside directors on firm value, *Journal of Business* 78, 2229–2254.
- Gilson, Stuart, 1990, Bankruptcy, boards, banks and blockholders: Evidence on changes on corporate ownership and control when firms default, *Journal of Financial Economics* 27, 355–387.
- Gompers, Paul, Joy Ishii, and Andrew Metrick, 2003, Corporate governance and equity prices, *Quarterly Journal of Economics* 118, 107–155.
- Harford, Jarrad, 2003, Takeover bids and target directors' incentives: The impact on directors wealth and board seats, *Journal of Financial Economics* 69, 51–83.
- Hermalin, Benjamin, and Michael Weisbach, 1988, The determinants of board composition, *Rand Journal of Economics* 19, 589–606.
- Hermalin, Benjamin, and Michael Weisbach, 1998, Endogenously chosen boards of directors and their monitoring of the CEO, *American Economic Review* 88, 96–118.
- Huson, Mark, Robert Parrino, and Laura Starks, 2001, Internal monitoring mechanisms and CEO Turnover: A long-term perspective, *Journal of Finance* 56, 2265–2297.

- Kaplan, Steven, and David Reishus, 1990, Outside directorships and corporate performance, Journal of Financial Economics 27, 389–410.
- Mace, Myles, 1986, Directors: Myth and Reality (Harvard University Press, Boston, MA).
- Mehran, Hamid, and David Yermack, 1997, Compensation and top management turnover, Working paper, New York University, Stern School of Business.
- Merton, Robert C., 1973, Theory of rational option pricing, Bell Journal of Economics and Management Science 4, 141–183.
- Parrino, Robert, 1997, CEO turnover and outside succession: A cross-sectional analysis, *Journal of Financial Economics* 46, 165–197.
- Parrino, Robert, Richard Sias, and Laura Starks, 2003, Voting with their feet: Institutional ownership changes around forced CEO turnover, *Journal of Financial Economics* 68, 3–46.
- Perfect, Steven, and Kenneth Wiles, 1994, Alternative constructions of Tobin's Q: An empirical comparison, *Journal of Empirical Finance* 1, 313–341.
- Perry, Tod, and Urs Peyer, 2005, Board seat accumulation by executives: A shareholder's perspective, *Journal of Finance* 60, 2083–2123.
- Rosenstein, Stuart, and Jeffrey Wyatt, 1990, Outside directors, board independence, and share-holder wealth, *Journal of Financial Economics* 26, 175–191.
- Shin, Hyun-Han, and Rene Stulz, 2000, Firm value, risk, and growth opportunities, Working paper, National Bureau of Economic Research 7808.
- Shivdasani, Anil, 1993, Board composition, ownership structure and hostile takeovers, *Journal of Accounting and Economics* 16, 167–198.
- Shivdasani, Anil, and David Yermack, 1999, CEO involvement in the selection of new board members: An empirical analysis, *Journal of Finance* 54, 1829–1853.
- Smith, Clifford, and Ross Watts, 1992, The investment opportunity set and corporate financing, dividend and compensation policies, *Journal of Financial Economics* 40, 263–292.
- Vafeas, Nikos, 1999, Board meeting frequency and firm performance, *Journal of Financial Economics* 53, 113–142.
- Weisbach, Michael, 1988, Outside directors and CEO turnover, *Journal of Financial Economics* 20, 431–460.
- White, Halbert, 1980, A heteroskedasticity-consistent covariance matrix estimator and a direct test for heteroskedasticity, *Econometrica* 48, 431–460.
- Yermack, David, 1996, Higher market valuation of companies with a small board of directors, Journal of Financial Economics 40, 185–211.
- Yermack, David, 2004, Remuneration, retention, and reputation incentives for outside directors, Journal of Finance 59, 2281–2308.

Copyright of Journal of Finance is the property of Blackwell Publishing Limited and its content may not be copied or emailed to multiple sites or posted to a listserv without the copyright holder's express written permission. However, users may print, download, or email articles for individual use.